Author's response to reviews

Title: OPAL: Randomized, controlled clinical trial to investigate oxygen persufflation as adjunct in liver preservation

Authors:

Thomas Minor (tminor@uni-bonn.de)
Carolin Pütter (Carolin.puetter@uk-essen.de)
Anja Gallinat (a.gallinat@gmx.de)
Claudia Ose (Claudia.Ose@uk-essen.de)
Gernot Kaiser (Gernot.Kaiser@uk-essen.de)
Andre Scherag (Andre.Scherag@uk-essen.de)
Jürgen Treckmann (j.treckmann@gmx.net)
Andreas Paul (Andreas.Paul@uk-essen.de)

Version: 2 Date: 12 July 2011

Author's response to reviews:

the trial registry is now listed at the end of the abstract, along with the unique identifying number: Trial registration: Current Controlled Trials ISRCTN00167887